Author's response to reviews

Title: 3066 consecutive Gamma Nails. 12 years experience at a single centre.

Authors:

Alicja J Bojan (alicja.bojan@vgregion.se)
Claudia Beimel (claudia.beimel@stryker.com)
Andreas Speitling (andreas.speitling@stryker.com)
Gilbert Taglang (gtaglang@wanadoo.fr)
Carl Ekholm (carl.ekholm@vgregion.se)
Anders Joensson (anders.joensson@stryker.com)

Version: 6 Date: 1 June 2010

Author's response to reviews: see over
Gothenburg, 30th May, 2010

**Re: 3066 consecutive Gamma Nails. 12 years experience at a single centre.**

To the Editor and the Reviewers,

Thank you for the comments.

We have added in Conclusions “Further evaluation of hip fracture treatment should involve multi-centre randomised prospective trials comparing cephalo-medullary nails and sliding hip screws.” as an answer to the Reviewer’s question

According to the Reviewer’s suggestion, we have changed the expression: “Left- and right-sided fractures were equally common” to “Left-sided fractures were present in 50.7% of the patients.”

The main text, figures and tables have been formatted according to BMC templates.

The institutional review board at CTO hospital gave ethics approval before the study was commenced. Due to the retrospective nature of the study no burden or risk was imposed on the patients. This comment has been included in the Methods section.

I would be grateful if you could send the submission fee to the following address:

Stryker Trauma GmbH,
Prof.-Kuentscher-Str. 1-5
24 232 Schönkirchen,
Germany

Yours sincerely,

Alicja Bojan, MD